Author's response to reviews

Title: Takotsubo cardiomyopathy in a Caucasian Italian woman: Case report

Authors:

Matteo Lisi (m.lisi@libero.it)
Valerio Zaca (valezeta@libero.it)
Francesca Casucci (f.casucci@yahoo.it)
Marianna Maggi (mmaggi@tin.it)
Stefano Lunghetti (s.lunghetti@libero.it)
Paolo Aitiani (P.Aitiani@ao-siena.toscana.it)
Arcangelo Carrera (A.Carrera@ao-siena.toscana.it)
Debora Castellani (castellanid@unisi.it)
Roberto Favilli (R.Favilli@ao-siena.toscana.it)
Carlo Pierli (C.Pierli@ao-siena.toscana.it)
Sergio Mondillo (mondillo@unisi.it)

Version: 3 Date: 2 April 2007

Author's response to reviews: see over
April 1, 2007

Dear editor:

We are submitting the revised version of the attached contribution entitled “Takotsubo cardiomyopathy: the importance of being stressed” for consideration for publication in *Cardiovascular Ultrasound* as a Case Report. In our manuscript we report the case of a postmenopausal woman who suffered the left ventricular transient apical ballooning syndrome shortly after an acute emotional stress. The manuscript has not been published elsewhere and is not under consideration for publication elsewhere. All the authors have read the revised version of the manuscript and approved its submission to *Cardiovascular Ultrasound*. We hope that you, the Section Editor and the Associate Editors assigned to this revised manuscript will find it suitable for publication in *Cardiovascular Ultrasound*.

Sincerely,

Valerio Zacà, MD
Department of Cardiology
University of Siena
ITALY